Author’s response to reviews

Title: Progressive stenosis and radiological findings of vasculitis over the entire internal carotid artery in moyamoya vasculopathy associated with Graves’ disease: A case report and review of the literature

Authors:

Hiroto Ito (hiroto.ito.nagoya.uni.med@gmail.com)
Syunsuke Yokoi (yokoisyunsuke0619@gmail.com)
Kinya Yokoyama (swallow_tale@hotmail.com)
Takumi Asai (takuchan47@hotmail.com)
Kenji Uda (kenji84u@med.nagoya-u.ac.jp)
Yoshio Araki (y.araki@med.nagoya-u.ac.jp)
Syuntaro Takasu (syun@nagoya2.jrc.or.jp)
Rei Kobayashi (reireik@nnh.hosp.go.jp)
Hisashi Okada (kada-q@nnh.hosp.go.jp)
Satoshi Okuda (okudas@nnh.hosp.go.jp)

Version: 1 Date: 18 Sep 2018

Author’s response to reviews:

We would like to thank the editor for the valuable and helpful comments on our initial manuscript. We appreciate this opportunity to revise our manuscript based on the editor’s recommendations. Our point-by-point responses to the editor comments follow.

1. Submission Guidelines:

Please ensure that your abstract and main manuscript follows our submission guidelines. For more information, please see: https://bmcneurol.biomedcentral.com/submission-guidelines/preparing-your-manuscript/case-report

We are thankful for this suggestion. We checked your submission guideline again. Then, we changed the subheadings in our abstract and our manuscript as the editor recommended. We
moved positions of the “Methods” section and “Review of the literature” section in our manuscript to Discussion.

2. Tables:

Please only include one copy of tables.

We appreciate this suggestion. We removed one copy of tables.

3. Supplementary Figures:

Please ensure that the supplementary figures have been uploaded in the system as supplementary material and appear in the correct order.

Thank you for this beneficial suggestion. We uploaded supplementary figures as supplementary material in the correct order.

4. Authors' Contributions:

Please note that based on the current authors’ contributions section, several listed authors do not automatically qualify for authorship.

We ask that you ensure that author contributions are in line with the ICMJE guidelines (below), and that all listed authors have performed all four points specified below.

We are thankful for this suggestion. We ensured that author contributions are in line with the ICMJE guidelines. We changed expression of our authors' contribution in more details.